

Original Paper

Relationship Between Caregiver Burden and Cognitive **Impairment in Adult Patients with Type 2 Diabetes**





Sorur Javanmardifard¹ , Fatemeh Shirazi² , Sheida Jamalnia^{3*} , Erfan Sadeghi⁴

- 1. Nursing (PhD), Instructor, School of Nursing and Midwifery, Shiraz University of Medical Sciences, Shiraz, Iran.
- 2. Assistant Professor, Community Based Psychiatric Care Research Center, School of Nursing and Midwifery, Shiraz University of Medical Sciences, Shiraz, Iran.
- 3. Medical Journalism (MSc), School of Paramedical sciences, Shiraz University of Medical Sciences, Shiraz, Iran.
- 4. Biostatistics and Epidemiology (Ph.D), Research Consultation Center (RCC), Shiraz University of Medical Sciences, Shiraz, Iran.



Citation Javanmardifard S, Shirazi F, Jamalnia Sh, Sadeghi E. Relationship Between Caregiver Burden and Cognitive Impairment in Adult Patients with Type 2 Diabetes. J Holist Nurs Midwifery. 2022; 32(3):203-209. https://doi.org/10.32598/jhnm.32.3.2207 Running Title Caregiver Burden and Cognitive Impairment in Diabetes



doi https://doi.org/10.32598/jhnm.32.3.2207



Article info:

Received: 01/03/2021 Accepted: 23/01/2022 Available Online: 01/07/2022

Keywords:

Caregiver burden, Cognitive function, Diabetes Mellitus

ABSTRACT

Introduction: Caregivers play a critical role in health management of patients with chronic diseases. Caregiving is associated with experiencing different emotional and psychological problems.

Objective: This study aims to determine the relationship between caregiver burden and cognitive function in adult patients with Type 2 Diabetes (T2D).

Materials and Methods: In this cross-sectional study, 215 family caregivers of adult patients with T2D were selected by a systematic random sampling method. A socio-demographic form, the Caregiver Burden Inventory (CBI), and the Mini-Mental State Examination (MMSE) were used for data collection. Data were analyzed using Kruskal-Wallis test, Spearman's rho correlation test. The multinomial logistic regression model was used to examine the power of Cognitive Impairment (CI) in predicting the caregiver burden. P<0.05 was considered as the statistically significance level.

Results: The mean age of caregivers and patients were 48±9 and 52±2 years, respectively. The mean score of CBI was 24.57±18.2, indicating a moderate level of caregiver burden. The mean score of MMSE was 25.25±3.6, indicating a mild CI in patients. Caregiver burden showed a significant correlation with cognitive function (r=-0.336, 95% CI; 0.53-0.89, P=0.001) such that the decreased CI was associated with an increase in total caregiver burden (r=-0.348, 95% CI; 0.43-0.87, P=0.001). The regression results showed that the CI levels were not predictors of caregiver burden severity.

Conclusion: The CI of patients with T2D may affect their caregivers' burden. Therefore, assessing the cognitive function of these patients can help healthcare providers conduct interventions to enhance the caregivers' quality of life.

Address: School of Paramedical sciences, Shiraz University of Medical Sciences, Shiraz, Iran.

Tel: +98 (933) 2721342

E-mail: Sh.jamalnia1988@gmail.com

Corresponding Author:



Highlights

- More than half of patients with type 2 diabetes had normal cognitive function to mild cognitive impairment.
- Most of caregivers reported moderate burden.
- Higher level of Cognitive Impairment (CI) in diabetic patients may cause an increase in the caregivers' burden.

Plain Language Summary

This study aimed to assess the relationship between caregiver burden and Cognitive Impairment (CI) in adult patients with Type 2 Diabetes (T2D). Many caregivers of these patients, particularly females suffer from psychological distress due to caregiving job. In order to enhance the quality of care for these patients, physicians should assess the burden of caregivers. Our results demonstrated that the caregivers had a moderate level of burden and the patients had mild cognitive impairment. Caregiver burden showed a significant correlation with cognitive impairment. Accordingly, increase in cognitive impairment was accompanied by an increase in the severity of caregiver burden. However, the cognitive impairment level were not the significant predictors of caregiver burden severity. It can be concluded that the cognitive function of patients with T2D may affect their caregivers' burden.

Introduction

iabetes is a costly disease due to causing various complications [1, 2]. The chronic nature of diabetes and its related complications are associated with both tangible and intangible costs for patients and their caregivers, and reduce the patients' quality of life [3, 4]. According to the International Diabetes Federation (IDF), three out of four individuals with diabetes (352 million people) are of working age (20-64 years of age). This figure is expected to increase to 417 million by 2030 and 486 million by 2045. In 2019, approximately 54.8 million people aged 20-79 years (12.8%) in the Middle East and North Africa were diagnosed with diabetes, along with a total of 24.5 million adults whose diabetes was undiagnosed. Pakistan (19.4 million), Egypt (8.9 million), and Iran (5.4 million) are the countries with the highest number of diabetic adults aged 20-79 years. In Iran, the prevalence of diabetes among adults aged 20-79 years in 2019 was reported 9.4% (ranged 7.4-12.3%) [5].

The effects of diabetes are not limited to early and late complications such as hyperglycemia and those related to the small and large arteries [6]. Diabetes can cause primary and secondary disorders as well as the central nervous system complications such as cognitive dysfunction [7]. Cognitive Impairment (CI) is one of the less known complications of Type 2 Diabetes (T2D) [8]. The prevalence of diabetes in people with CI ranges 6-39 % [9]. The CI includes reduced information processing speed, concentration, memory, learning ability,

problem-solving ability, visual intelligence, and mental flexibility [10]. It negatively influences self-care behaviors. Poor glycemic control has been shown to be a risk factor for CI [11]. Furthermore, diabetes can worsen cognitive aging, which is a risk factor for CI. Therefore, the caregivers should pay attention to the CI of diabetic patients [12].

The burden of care for diabetic patients is heavy due to the deteriorating nature of the disease, which may affect the caregivers' daily activities [12]. In caring for patients with diabetes, 66.7% and 22.5% of family caregivers experience objective and subjective family burdens, respectively [13, 14]. Caregiver burden is not only associated with negative emotional conditions and psychological disorders; it has other physical, financial, and social consequences. Compared to non-caregivers, caregivers are more likely to have poor health, seek medical treatment, and take psychotropic medicines more frequently [15]. The caregiver burden is also associated with negative consequences for the patients and is closely related to their institutionalization [16]. Caregivers with a previous history of psychiatric disorders are more prone to relapse [1]. Connors et al. reported high levels of caregiver burden in caregivers of people with mild CI [3]. Patients' higher levels of neuropsychiatric symptoms, decreased functional ability, and loss of driving abilities as well as the caregivers' occupational challenges are associated with increased caregiver burden over time [17]. Jane et al. reported that many Nigerian caregivers of patients with T2D experienced the burden and psychological distress as a result of caregiving tasks [18].



There is a need for more research in Iran on the impact of diabetes on caregivers. Most of the studies on caregiver burden have been conducted in Western countries. Since Asian countries have diverse cultures, the focus on the psychosocial aspects can be beneficial. The difference in cultures can lead to different results. According to studies, caregivers of T2D patients have heavy care burden, psychological distress, and psychological illnesses such as depression and anxiety [18, 19]. The present study aims to determine the relationship between caregiver burden and CI in adult patients with T2D.

Materials and Methods

In this cross-sectional study, 215 patients with T2D were selected from those referred to Shiraz Diabetes Association clinic from July 2017 to July 2018 using a random sampling method who met the eligibility criteria. The sample size was determined 215 based on Green's formula for regression sample size: N≥50 +8k [20] where k (the number of independent variables) was set at 20. However, to prevent possible dropout, the sample size was increased to 225. One in three patient/caregiver dyads who met the inclusion criteria were selected each day from the clinic during the study using systematic random sampling. Inclusion criteria for patients were age 25-65 years, diagnosis of T2D based on the 1999 World Health Organization's diagnostic criteria for diabetes, suffering from diabetes for more than a year, and having at least a primary school education. Inclusion criteria for caregivers were: Being a caregiver of patients with T2D, age ≥18 years, caregiving for at least 6 months, willingness to participate, being the patient's first-degree relatives (child, spouse, father, mother, sister, or brother), living with the patient, care for the patient solely because of diabetes or its complications, no history of mental disorders or participation in psychotherapy courses, and not suffering from disabling physical diseases due to taking care of the patient. The exclusion criteria were incomplete return of the questionnaire.

The data collection tool was a socio-demographic form, the Mini–Mental State Examination (MMSE), and the Caregiver Burden Inventory (CBI). The MMSE or Folstein test is a 30-point questionnaire used to assess CI in clinical and research settings. This instrument was developed by Folstein et al. for brief quantitative assessment of cognitive function. It can screen and assess the CI severity, and monitor the changes. Its items assess various cognitive areas, including orientation in time and place, repetition, verbal recall, attention and calculation, language, and visual construction. It takes 5–10 minutes to

complete. The total score ranges from 0 (impaired) to 30 (normal). Accordingly, scores 0-9, 10-20, 21-26, and 27-30 represented severe, moderate, mild CI, and normal cognitive function, respectively [21]. Using a cut-off score of <24, MMSE was found to be 87% sensitive and 82% specific in detecting dementia and delirium among hospital patients [22]. In a study by Foroughan and et al., results showed acceptable reliability and validity of the Persian version of MMSE [23]. The MMSE questionnaire was completed by the researcher for the patients.

The caregivers were assessed using the CBI. This diverse, multidimensional instrument was developed by Novak and Guest to measure the care burden in caregivers [24]. It has 24 items and 5 subscales. There are five items for each subscale, except for the physical subscale which contains four items: Time dependence (burden due to shortage of time; items 1-5), developmental (sense of failure in meeting role expectations and enjoying opportunities compared to the peers; items 6-10), physical (feeling of fatigue and chronic health problems; items 11-14), social (due to perceived conflict of roles; items 15-19), and emotional (due to negative feelings towards the patients due to their bizarre and unpredictable behaviors; items 20-24). The items were scored one a five-point Likert scale from 0 (not at all disruptive) to 4 (very disruptive). Thus, the total score of each subscale range from 0 to 20, except for the physical subscale which had four items. Therefore, the total CBI score ranges from 0 to 96, with higher scores representing higher burdens. Accordingly, scores 0-19, 20-50, and 51-96 represent normal to mild, mild to moderate, and moderate to severe burdens, respectively. The CBI takes approximately 10-15 minutes to complete. The Persian version of CBI which was localized by Shafizadeh et al. [25] were used in this study.

The continuous variables were described by mean and standard deviation, while the categorical variables were presented by percentage. Kolmogorov-Smirnov test was used to analyze the normality of data distribution. Difference between different CI levels were analyzed using Kruskal-Wallis Test. The correlation between the continuous variables were evaluated using Spearman's correlation test. The multinomial logistic regression model was used to examine the power of CI in predicting the caregiver burden after adjusting socio-demographic variables. All statistical analyses were performed in SPSS v. 24 software, and P<0.05 was considered as the statistically significance level.



Results

In this study, 10 caregivers did not complete their questionnaires. Thus, they were excluded from the study, and the research was carried out on 215 caregivers. The mean age of caregivers and patients were 48±9 and 52±2 years, respectively. The majority of caregivers were female (78.6%) and married (91.6%). Most of them were daughters (35.3%) and spouses (35.3%) of the patients, and had lower than high school education (47.9%). Moreover, T2D had presented in 65.1% of patients after the age of 60 years. The mean score of MMSE was 25.25±3.6 in the patients; 46.5% had normal cognitive function, 46.5% had mild CI, and 7% had moderate CI. Furthermore, most of the caregivers (45.8%) reported moderate burdens (Table 1). The mean score of CBI was 24.57±18.2.

A decrease in CI level was correlated with an increase in total level of caregiver burden (r=-0.348, P=0.001), and an increase in the severity of caregiver burden (r=-0.366, P=0.001) based on Spearman rho test. There was no significant difference between caregiver burden severity and CI levels (Table 2). The multinomial logistic regression results showed that the CI levels were not predictors of caregiver burden severity after adjusting demographic variables (Table 3).

Discussion

The present study indicated that most patients had a normal to mild CI. No severe CI case was detected. Moattari et al. in their study also found no T2D patient with severe CI; most of patients had normal to mild CI. Caregivers may experience a high level of burden when caring for T2D patients, which can seriously affect the patients' lives [26]. The number of patients with T2D and CI or dementia is expected to increase worldwide

Table 1. Demographic characteristics of caregivers (n=215)

Charac	No. (%)	
Gender	Male	46(21.4)
Gender	Female	169(78.6)
Marital status	Single	19(8.4)
iviai itai status	Married	196(91.6)
	Employed	83(38.6)
Occupation	Retired/not working	48(22.3)
	Housekeeper	84(39.1)
	Lower than high school education	103(47.9)
Educational level	Bachelor's degree or lower	93(43.3)
	Higher than bachelor's degree	19(8.8)
	Son	63(29.3)
Relationship with patient	Daughter	76(35.3)
	Spouse	76(35.3)
History of psychiatric disorders	No	186(86.5)
Thistory of psychiatric disorders	Yes	29(13.5)
	Low	97(45.3)
Caregiver burden severity	Mild	98(45.8)
	High	19(8.9)



Table 2. Caregiver burden based on Cognitive Impairment (CI) levels

	No. (%)/Mean±SD				
Caregiver Burden	CI level				
	Normal (n=100)	Mild (n=100)	Moderate (n=15)	Total (n=215)	
Low	46(46.9)	43(43.9)	9(9.2)	98(100)	
Mild	46(46.9)	47(48.0)	5(5.1)	98(100)	0.793
High	8(42.1)	10(52.6)	1(5.3)	19(100)	
Total Caregiver burden	23.68±17.58	25.48±18.09	24.40±23.21	24.57±18.18	0.711

^{*}Kruskal-Wallis Test

due to the prevalence of diabetes and the concomitant rise in adult populations. In this respect, dementia can be a new long-term diabetic complication with emotional consequences for the affected individuals and their families, and with a significant effect on healthcare providers. Therefore, more studies for recognizing T2D patients at risk of dementia are required.

The results of the present study showed a significant correlation between the caregiver burden and cognitive function. Higher level of CI in patients may increase the caregivers' burden. This implies that the caregiver burden can be decreased by modifying the patient's cognitive function. Similarly, Lara-Ruiz et al. found that the patients' neuropsychological scores predicted caregiver burden less than their daily functioning [15]. Caring for another person may be perceived as stressful job. Family caregivers spend emotional, financial, and physical costs to be constantly available to family members. Our

findings confirmed the presence of burden among the caregivers of T2D patients with CI, with a prevalence of about 36%. There are increased caregiver burden when patients become more disable and require more assistance with, for example, financial and medication management. In this study moreover increased caregiver burden is associated with a longer course of CI and greater cognitive, behavioral and emotional problems among the patients.

Our results showed a mild to moderate caregiver burden among the caregivers of diabetic patients with different CI levels. This burden may be due to the chronic nature of the disease, the patients' disability in self-care, the caregivers' increased responsibility, inadequate attention to the caregivers as a member of healthcare team, caregivers' lack of satisfaction, and socioeconomic factors. The caregivers need for rest, being alone, and being with friends may be neglected due to caregiving

Table 3. Multinomial logistic regression model for assessing the predictive role of Cognitive Impairment (CI) to Caregiver burden

CI	Caregiver Burden	Beta			95% CI		- P
			Standard Error	OR	Bound		
					Lower	Upper	
Mild	Low	-0.547	0.555	0.579	0.195	1.718	0.325
	Mild	-0.390	0.548	0.677	0.231	1.982	0.477
	High	-	-	-	-	-	-
Moderate	Low	0.497	1.171	1.644	0.166	16.324	0.671
	Mild	-0.128	1.202	0.880	0.083	9.280	0.915
	High	-	-	-	-	-	-

Note: normal level of cognitive impairment was considered as reference adjusted for age, gender, educational level, occupation, marital status, relationship with patient, disease history, and disease onset.



responsibilities. They have individual and social needs. With increase in the disease complications including CI, the patient may become more dependent on the caregiver, leading to more difficult caregiving job. Caring for a diabetic patient with various levels of CI has specific challenges [20]. Caro et al., however, demonstrated that the caregivers had moderate levels of burden [16]. This discrepancy in the results may be due to difference in the study populations and the instruments used.

Female caregivers, even those who are not mothers, are more common in Iran and receive less support and have less resting compared to male caregivers. This is also common among other low and middle-income countries, particularly among poor families. The social determinants can affect or hinder the successful caregiving for the patients [27]. In Iranian society and culture, women are primarily selected for taking care of patients and the elderly. Studies have also reported that females make up the majority of family caregivers [3, 17].

Given that in this study, a questionnaire was used for assessing caregiver burden which may not reflect the complete caregiver burden. Other methods are required to recognize the risk factors among caregivers, and planning should be done for providing support to them. On the other hand, hiring a professional nurse can place less burden on caregivers. Health care providers and policymakers should pay more attention to this important issue and take proper measures in the field of care for diabetic patients to reduce the burden of care in families' members.

There is a moderate caregiver burden among the caregivers of T2D patients with CI. Steps should be taken to reduce burden among the caregivers to enable them to provide the best healthcare services to the patients. Otherwise, they may experience problems which can disturb their caregiving role.

Ethical Considerations

Compliance with ethical guidelines

This research received ethical approval from the ethics committee of Shiraz University of Medical Sciences (Code: IR.SUMS.REC.1397.262). Written informed consent was obtained from the participants.

Funding

This study was funded by Shiraz University of Medical Sciences.

Authors' contributions

All authors equally contributed to preparing this article.

Conflict of interest

The authors declare that they have no conflict of interest.

Acknowledgments

The authors would like to thank Shiraz University of Medical Sciences for their financial support, and all diabetic patients and their caregivers for their cooperation.

References

- [1] Jones AJ, Kuijer RG, Livingston L, Myall D, Horne K, MacAskill M, et al. Caregiver burden is increased in Parkinson's disease with mild cognitive impairment (PD-MCI). Translational Neurodegeneration. 2017; 6:17. [PMID] [PMCID]
- [2] Paradise M, McCade D, Hickie IB, Diamond K, Lewis SJ, Naismith SL. Caregiver burden in mild cognitive impairment. Aging & Mental Health. 2015; 19(1):72-8. [PMID]
- [3] Connors MH, Seeher K, Teixeira-Pinto A, Woodward M, Ames D, Brodaty H. Mild cognitive impairment and caregiver burden: A 3-year-longitudinal study. The American Journal of Geriatric Psychiatry. 2019; 27(11):1206-15. [PMID]
- [4] George C, Ferreira N. Carer burden and positive gain in dementia. Working with Older People. 2020; 24(2):113-23. [DOI:10.1108/ WWOP-02-2020-0009]
- [5] American Diabetes Association. 2. Classification and diagnosis of diabetes: Standards of medical care in diabetes-2019. Diabetes Care. 2019; 42(Supplement 1):S13-28. [PMID]
- [6] Baxter M, Hudson R, Mahon J, Bartlett C, Samyshkin Y, Alexiou D, et al. Estimating the impact of better management of glycaemic control in adults with Type 1 and Type 2 diabetes on the number of clinical complications and the associated financial benefit. Diabetic Medicine. 2016; 33(11):1575-81. [PMID]
- [7] Biessels GJ, Strachan MW, Visseren FL, Kappelle LJ, Whitmer RA. Dementia and cognitive decline in type 2 diabetes and prediabetic stages: towards targeted interventions. The lancet Diabetes & Endocrinology. 2014; 2(3):246-55. [DOI:10.1016/S2213-8587(13)70088-3]
- [8] Simó R, Ciudin A, Simó-Servat O, Hernández C. Cognitive impairment and dementia: A new emerging complication of type 2 diabetes-the diabetologist's perspective. Acta Diabetologica. 2017; 54(5):417-24. [PMID]
- [9] Bunn F, Burn AM, Goodman C, Rait G, Norton S, Robinson L, et al. Comorbidity and dementia: A scoping review of the literature. BMC Medicine. 2014; 12:192. [PMID] [PMCID]
- [10] Saedi E, Gheini MR, Faiz F, Arami MA. Diabetes mellitus and cognitive impairments. World Journal of Diabetes. 2016; 7(17):412-22. [PMID] [PMCID]



- [11] Albai O, Frandes M, Timar R, Roman D, Timar B. Risk factors for developing dementia in type 2 diabetes mellitus patients with mild cognitive impairment. Neuropsychiatric Disease and Treatment. 2019; 15:167-75. [PMID] [PMCID]
- [12] Sheikh F, Ismail Z, Mortby ME, Barber P, Cieslak A, Fischer K, et al. Prevalence of mild behavioral impairment in mild cognitive impairment and subjective cognitive decline, and its association with caregiver burden. International Psychogeriatrics. 2018; 30(2):233-44. [PMID]
- [13] Given BA, Given CW, Sherwood PR. Family and caregiver needs over the course of the cancer trajectory. The Journal of Supportive Oncology. 2012; 10(2):57-64. [PMID]
- [14] Kristianingrum ND, Ramadhani DA, Hayati YS, Setyoadi S. Correlation between the burden of family caregivers and health status of people with diabetes mellitus. Journal of Public Health Research. 2021; 10(2):2227. [DOI:10.4081/jphr.2021.2227] [PMID] [PMCID]
- [15] Lara-Ruiz J, Kauzor K, Nakhala M, Banuelos D, Woo E, Apostolova LG, et al. The functional ability of MCI and Alzheimer's patients predicts caregiver burden. GeroPsych. 2019; 32(1):31-9. [PMID] [PMCID]
- [16] Caro CC, Mendes PV, Costa JD, Nock LJ, Cruz DM. Independence and cognition post-stroke and its relationship to burden and quality of life of family caregivers. Topics in Stroke Rehabilitation. 2017; 24(3):194-9. [PMID]
- [17] Kelleher M, Tolea MI, Galvin JE. Anosognosia increases caregiver burden in mild cognitive impairment. International Journal of Geriatric Psychiatry. 2016; 31(7):799-808. [PMID] [PMCID]
- [18] Ogunmodede AJ, Abiodun O, Makanjuola AB, Olarinoye JK, Ogunmodede JA, Buhari OI. Burden of care and psychological distress in primary caregivers of patients with type-2 diabetes mellitus in a tertiary hospital in Nigeria. Ethiopian Journal of Health Sciences. 2019; 29(6):697-708. [PMID] [PMCID]
- [19] Świątoniowska-Lonc N, Polański J, Tański W, Jankowska-Polańska B. Impact of cognitive impairment on adherence to treatment and self-care in patients with type 2 diabetes mellitus. Diabetes, Metabolic Syndrome and Obesity: Targets and Therapy. 2021; 14:193-203. [PMID] [PMCID]
- [20] Jamalnia S, Javanmardifard S, Akbari H, Sadeghi E, Bijani M. Association between cognitive impairment and blood pressure among patients with type ii diabetes mellitus in Southern Iran. Diabetes, Metabolic Syndrome and Obesity: Targets and Therapy. 2020; 13:289-96. [PMID] [PMCID]
- [21] Folstein MF, Folstein SE, McHugh PR. "Mini-mental state": A practical method for grading the cognitive state of patients for the clinician. Journal of Psychiatric Research. 1975; 12(3):189-98. [DOI:10.1016/0022-3956(75)90026-6]
- [22] Anthony JC, LeResche L, Niaz U, Von Korff MR, Folstein MF. Limits of the 'Mini-Mental State'as a screening test for dementia and delirium among hospital patients. Psychological Medicine. 1982; 12(2):397-408. [PMID]
- [23] Foroughan M, Jafari Z, Shirin BP, Ghaem Z, Rahgozar M. [Validation of mini-mental state examination (MMSE) in the elderly population of Tehran (Persian0]. 2008; 10(20):29-37. https://www.sid.ir/En/Journal/ViewPaper.aspx?ID=147028
- [24] Novak M, Guest C. Application of a multidimensional caregiver burden inventory. The Gerontologist. 1989; 29(6):798-803. [PMID]

- [25] Shafiezadeh A, Heravi-Karimoo M, Rejeh N, Sharif Nia H, Montazeri A. [Translation and primarily validation of the Persian Version of Caregiver Burden Inventory (Persian)]. Payesh (Health Monitor). 2019; 18(4):405-14. http://payeshjournal.ir/article-1-1100-en.html
- [26] Moattari M, Jamalnia S, Mansoori P, Mani A, Dabbaghmanesh MH, Sayadi M. The prevalence of cognitive impairment in Patients with Type II Diabetes and its Relationship with Quality of Life, Self-Management Profiles, and HbA1c. Iranian Journal of Diabetes and Obesity. 2016; 8(1):35-40. http://ijdo.ssu.ac.ir/article-1-283-en. html
- [27] Jamalnia S, Javanmardifard S. A review on relationship between type ii diabetes and cognitive impairment. Iranian Journal of Diabetes and Obesity. 2018; 10(3):159-63. http://ijdo.ssu.ac.ir/article-1-432-en.html